



# **JGR** Biogeosciences

# **RESEARCH ARTICLE**

10.1029/2023JG007931

#### **Key Points:**

- Contribution of different drivers to N runoff loss was compared between hills and plains
- Fertilization contributed more to N loss in hills due to N increment of paddy land
- Rainfall contributed more to N loss in plains due to N increment of urban land

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

Q. Chen and C. Chen, qwchen@nhri.cn; chencheng@nhri.cn

#### Citation:

Cui, Z., Chen, C., Chen, Q., & Huang, J. (2024). Difference in the contribution of driving factors to nitrogen loss with surface runoff between the hill and plain agricultural watersheds. *Journal of Geophysical Research: Biogeosciences*, *129*, e2023/G007931. https://doi.org/10. 1029/2023JG007931

Received 3 DEC 2023 Accepted 15 FEB 2024

#### **Author Contributions:**

Conceptualization: Zhen Cui Data curation: Zhen Cui, Jiacong Huang Formal analysis: Zhen Cui Funding acquisition: Qiuwen Chen Investigation: Zhen Cui Methodology: Zhen Cui, Cheng Chen Software: Jiacong Huang Supervision: Qiuwen Chen Validation: Cheng Chen Visualization: Zhen Cui, Cheng Chen Writing – original draft: Zhen Cui Writing – review & editing: Qiuwen Chen

© 2024. American Geophysical Union. All Rights Reserved.

# Difference in the Contribution of Driving Factors to Nitrogen Loss With Surface Runoff Between the Hill and Plain Agricultural Watersheds

Zhen Cui<sup>1,2</sup>, Cheng Chen<sup>1,2,3</sup>, Qiuwen Chen<sup>1,2,4</sup>, and Jiacong Huang<sup>5</sup>

<sup>1</sup>The National Key Laboratory of Water Disaster Prevention, Nanjing Hydraulic Research Institute, Nanjing, China, <sup>2</sup>Center for Eco-Environmental Research, Nanjing Hydraulic Research Institute, Nanjing, China, <sup>3</sup>College of Water Conservancy and Hydroelectric Power, Hohai University, Nanjing, China, <sup>4</sup>Yangtze Institute for Conservation and Green Development, Nanjing, China, <sup>5</sup>Key Laboratory of Watershed Geographic Sciences, Nanjing Institute of Geography and Limnology, Chinese Academy of Sciences, Nanjing, China

**Abstract** Identifying the factors and quantifying their contributions to nitrogen (N) loss associated with surface runoff is of great significance to the control of non-point source N pollution. However, the distinct geographical units, such as hills and plains, may lead to great differences in the contribution of these driving factors, which has been rarely investigated. This study developed an effective framework, which simulated the N loss with surface runoff in hills and plains by SWAT and NDP, and analyzed their spatial distribution variations by spatial autocorrelation analysis, and distinguished the contribution of their driving factors by multi-scenario simulation and partial redundancy analysis (pRDA). The framework was instantiated in a hill and a plain agricultural watershed, respectively, in the upper Taihu Lake Basin, China. We found the contribution of fertilization to N loss with surface runoff in the hills (10.4%) was greater than that in the plains (6.4%), which may be due to the N increment effect of paddy land. The contribution of rainfall to N loss with surface runoff in the hills (74.3%), which may be due to the N increment effect of urban land. The developed framework could provide a viable way to study the environmental impacts of natural and anthropogenic drivers in different types of agricultural watersheds, thus offer scientific references for nutrient control measures.

**Plain Language Summary** Geographical units within a watershed, such as hills and plains, exhibit distinct characteristics that give rise to variations in the processes of nitrogen (N) runoff loss and their associated driving factors. These variations pose challenges to the development of effective control measures for non-point source N pollution. Consequently, we have developed an effective framework capable of simulating N loss with surface runoff, analyzing variations in their spatial distribution, and discerning the external (e.g., rainfall) and internal (e.g., land management, fertilization) contributions of driving factors in the hills and plains, respectively. A notable discovery was that fertilization played a more substantial role in N loss in hills due to

increased N levels resulting from paddy land, whereas rainfall emerged as the primary driver of N loss in plains due to elevated N levels associated with urban land. This research contributes to our understanding of the complexities surrounding N runoff loss and provides valuable insights for the development of targeted control measures within diverse geographical units.

#### 1. Introduction

With the growth of population and the continuous advancement of agriculture and urbanization, land has been highly utilized and fertilizer has been heavily invested to ensure food production. Excessive nitrogen (N) eventually enters to the surrounding water bodies mainly through surface runoff (including drainage), which poses a hidden danger to the sustainable development of water environment and the safety of agricultural products (Zhang, Hou, et al., 2020). On a global basis, humans have more than doubled the amount of N fixation, with highly industrialized and agriculturalized areas experiencing mineralized N concentration up to 25 times that of pre-development concentrations (Alldred & Baines, 2016). Agricultural non-point source N loss accounts for 80%–90% of the total N pollution in water bodies, which mainly occurs in the rainy season (Law et al., 2022; Zhang, Hou, et al., 2020). Therefore, identifying the spatial-temporal dynamics of N loss with surface runoff and its driving factors and contribution will provide a scientific basis for the fine management of watershed N regulation (Vystavna et al., 2023).



Much research has already been undertaken on the driving factors of N loss. For example, Gabriel et al. (2018) found that extreme rainfall or drought was expected to exacerbate current stresses on N loss from population and economic growth, land use and fertilization. However, land use has a more direct impact on N loss than climate through changing N sources and transport processes. In agricultural watersheds, rainfall affects agricultural production systems and also annual and seasonal variability in hydrologic processes, and thereby influences N runoff loss and its spatial distribution (Li et al., 2022; Wade et al., 2022; Wenng et al., 2020). Agricultural land generally produces the largest N load, followed by tea plantations, citrus groves and woodlands (Zhang, Li, et al., 2020). Although urban land accounts for a small proportion in agricultural watersheds, it increases the proportion of impervious surface, reduces land permeability, leads to higher runoff amount and non-point source N load especially during rainy seasons (Alamdari et al., 2022; Vystavna et al., 2023; Zhu et al., 2022). N fertilization can increase crop yield, but cannot be fully absorbed by crops, and then partially enter rivers and lakes through runoff or seepage. In summary, N runoff loss in agricultural watersheds varies among different land use and cropping patterns based on rainfall, surface vegetation, soil property and agricultural management (Ding et al., 2022).

Owing to the gaps in developmental status of each region, the changing trends in land use and fertilization vary from region to region. Moreover, uncertainties exist since it is not known how management practices may adapt to land use and climate change (Wenng et al., 2020). To assess the response of non-point source N pollution discharge to external (e.g., climate) and internal (e.g., water management, land management), the simulation evaluation method based on scenarios analysis was used in different watersheds, including the recent consideration of detailed N transport processes in small (<50 km<sup>2</sup>) watersheds and regional and global-scale assessments of total N (TN) load using empirical and process-based models (Alamdari et al., 2022). Hills and plains are the most common and representative landform types in a region, and their hydrological and N transport processes are significantly different. In the hills, the dry and wet conditions change dramatically, which makes it difficult to study the spatial and temporal pattern of water yield and N loss load (Bales et al., 2006; de Jong et al., 2005). The Soil Water Assessment Tool (SWAT) model is one of the best available tools for simulating the response of hill agricultural watersheds to water quality. Majority of studies conducted by the SWAT model focus on how to reduce N pollution through land use changes, as well as regulating fertilizer application rates (Oduor et al., 2023). However, the use of watershed models such as SWAT to simulate the hydrological and N transport processes in the plains is not ideal. The main reason is that the distributed watershed models divide the space into hydrological response units (HRUs), and determines the flow direction according to the elevation difference between these units, while the flow direction in the plains is more susceptible to water level difference. Moreover, most watershed models cannot track the artificial drainage processes such as pumping and culvert drainage, which control the hydrological and material exchange in the plains (Huang et al., 2018). Therefore, the conclusion of performances of driving factors with different watershed characteristics on N loss is still limited.

Aiming at the above knowledge gaps, we proposed a hypothesis: different from the great impact of fertilization in paddy land in the hill agricultural watershed, the impact of urbanization on N loss with surface runoff would be greater in plain agricultural watershed. To test this hypothesis, we developed a model framework for quantifying the N loss with surface runoff in hill and plain agricultural watersheds, and distinguishing the contribution of driving factors to spatial distribution variations of N loss.

## 2. Framework Development

The framework consists of several major steps as follows: (a) modeling the N loss and runoff processes in the hill agricultural watershed by SWAT model; (b) modeling the N loss and drainage processes in the plain agricultural watershed by NDP model; (c) analyzing the spatial distribution variation of N loss with surface runoff by spatial autocorrelation analysis; (d) quantifying the contribution of rainfall, land use and fertilization to N loss with surface runoff by multi-scenario simulation and partial redundancy analysis (pRDA) (Figure 1).

#### 2.1. Modeling in the Hill Agricultural Watershed

SWAT (Soil and Water Assessment Tool) model was used to simulate N loss with surface runoff in the hill agricultural watershed (Arnold, 1994). The hydrological processes described by the SWAT model were divided into the land surface part (i.e., runoff generation and overland flow) and the water surface part (i.e., flow routing in river channel) of the water cycle. The former determined the amount of water, sand, nutrients and chemicals input





**Figure 1.** Model framework for identifying the contribution differences of driving factors to N loss with surface runoff between the hill and plain agricultural watersheds.

from each sub-watershed to the main channel; the latter controlled the transport of material from the river network to the outlet of the watershed. Water yield was the total amount of water (i.e., surface runoff, lateral flow, groundwater) leaving the HRUs and entering main channel during the time step. SWAT contained N storage of ammonium N, nitrate N and organic N in the soil. Since land use was reflected at the HRU level, N load of different land use types was extracted from the HRUs. The TN loss load was calculated by multiplying various forms of the N concentration by the water amount in different hydrological pathways (Equation 1, Malago et al., 2017).

$$\Delta T N^T = \Delta T N^T_{Surf} + \Delta T N^T_{Lat} + \Delta T N^T_{GW} \tag{1}$$

where  $\Delta TN^T$  was the TN loss load (ton);  $\Delta TN_{Surf}^T$  was the TN loss load due to surface runoff (ton);  $\Delta TN_{Lat}^T$  was the TN loss load due to lateral flow (ton);  $\Delta TN_{GW}^T$  was the TN loss load due to groundwater (ton).

#### 2.2. Modeling in the Plain Agricultural Watershed

NDP (Nitrogen Dynamic Polder) model was used to simulate N loss with surface runoff (i.e., drainage) in the plain agricultural watershed (Huang et al., 2018). NDP driven by water level was used to realize the precise characterization of multi-scale and multi-process of non-point source N migration and transformation in the polders. It included five water balance modules and three N dynamic modules using a daily time step. The key augmentations of NDP model involved the representation of artificial drainage, as well as the water interactions among surface water, groundwater, and soil water in farmlands. These two mechanisms were particularly important due to their significant impact on water and N balance in the polders (Equation 2).

$$\Delta T N_{Polder}^T = \Delta T N_{Pump}^T + \Delta T N_{Culvert}^T + \Delta T N_{Seep}^T - \Delta T N_{Irr}^T$$
(2)

where  $\Delta T N_{Polder}^T$  was the TN loss load from the polder (ton);  $\Delta T N_{Pump}^T$  was the TN loss load due to flood drainage (ton);  $\Delta T N_{Culvert}^T$  was the TN loss load due to culvert drainage (ton);  $\Delta T N_{Seep}^T$  was the TN loss load due to seepage (ton);  $\Delta T N_{Tr}^T$  was the TN import amount due to irrigation (ton).

#### 2.3. Analyzing the Spatial Distribution Variations

The spatial autocorrelation analysis was used to study the spatial distribution of the classification results by TN concentration and to test whether the attribute value of a unit was related to the attribute value of its adjacent spatial units. The global spatial autocorrelation was performed to summarize the degree of spatial dependence in a total spatial range, which was expressed by Moran's I index; a clearly positive Moran's I indicated that the observed values of factors tend to be spatially aggregated; a clearly negative Moran's I indicated that the observed values of factors tend to be spatially dispersed. Local spatial autocorrelation was thereafter performed to describe the similarity between a spatial unit and its domain. It could indicate the extent to which each local unit followed a global general trend (including direction and magnitude), illustrating how spatial dependence changed with position (Zhao et al., 2019).

The bivariate Local Indicators of Spatial Association (LISA) was used as a straightforward extension of the LISA functionality to two different variables, one for the location and another for the average of its neighbors (Song et al., 2020). In this study, surface runoff amount or the proportion of agricultural land was selected as the x-variable and TN concentration was selected as the y-variable. A clearly positive Moran's I indicated significant positive spatial correlation and a clearly negative Moran's I indicated significant negative spatial correlation. All spatial changes of various factors were analyzed using GeoDa (Anselin, 2004) and ArcGIS 10.7.







# 2.4. Quantifying the Contribution of Driving Factors

To explore whether land use/management changes have a positive or negative impact on N loss, we examined three land use conversion scenarios (from 20% of paddy land to dry land, urban land and ponds, respectively) and two land management scenarios (reducing 20% and 40% of N fertilizer amount). The land management scenarios setting was based on some studies that proved reducing 20%–40% of N fertilizer amount in the rainy season did not influence the crop production (Xue et al., 2011).

To assess how much of the N loss can be explained by the variations of rainfall, land use and fertilization, the partial redundancy analysis (pRDA) performed by CANOCO 5.0 (ter Braak & Šmilauer, 2012) was used to quantify the unique and shared contributions of above three groups of environmental factors to the total variance. Annual accumulated rainfall, fertilizer amount and the proportion of agricultural land/urban land were selected as the *x*-variables, annual TN load was selected as the *y*-variable. The shared contribution was obtained by subtracting the unique variation of each explanatory variable from the total variation explained by all explanatory variables (Dalu et al., 2017).

# 3. Framework Application

#### 3.1. Study Area

The Xitiaoxi River Basin is located in the upper reaches of the Taihu Lake with advanced economy and dense population (Figure 2). The average fertilization amount is  $353 \text{ kg}\cdot\text{ha}^{-1}$ , exceeding the international ecological security standard with 225 kg·ha<sup>-1</sup>. As one of the most important tributaries of Taihu Lake, the Xitiaoxi River is 159 km long, suppling  $26.8 \times 10^8 \text{ m}^3$  (27.7%) of water volume to Taihu Lake, furthermore contributing greatly to the nutrient load (Li et al., 2017; Zhou et al., 2013). The terrain is high in the southwest and low in the northeast, with the elevation between 0 and 1,576 m. The area of the hill agricultural watershed (119°14′-119°48′E, 30°23′- 30°45′N) is 1,367 km<sup>2</sup>. Paddy land, dry land, pond and urban land account for 12%, 13%, 0.1% and 3% of the total area, respectively. The area of the plain agricultural watershed (119°48′-120°9′E, 30°45′-31°5′N) is 869 km<sup>2</sup>, which is composed by basic geographical units of 141 polders. The main land use types are paddy land (69%), followed by dry land (14%), urban land (11%) and pond (6%).

The study area belongs to subtropical monsoon climate, with an average annual temperature of  $15.5^{\circ}$ C and an average annual precipitation of 1,465.8 mm. More than 75% of rainfall occurs in the rainy season (from April to October), which is easily coincident with fertilizer, causing a large amount of farmland N runoff loss (Yuan et al., 2023). In the hill agricultural watershed, runoff is characterized by high intensity and short duration due to their topography. In the plain agricultural watershed, the pumps and ditches reduce the flood peak flow and prolong the hydraulic retention time, providing convenient conditions for the diffusion of N in the agricultural land (Huang et al., 2018).

According to the "Huzhou Water Resources Bulletin", the average annual precipitation in 2013 was 1,220.2 mm, which belonged to the dry year (12.7% less than the normal year); from July to August, there was continuous high temperature and little rain; residents in hills had difficulty in drinking water, meanwhile large areas of crops and freshwater aquaculture in plains suffered from drought. In 2015, the average annual precipitation was 1,675.1 mm, which belonged to the wet year (20% more than the normal year); the Mei-yu period lasted for 35 days, with the rainfall was 55.7% more than the annual; the average water yield coefficient of the watershed was 0.62. The average precipitation in 2017 was 1,334.1 mm, which belonged to the normal year; the average water yield coefficient was 0.48. Therefore, this study selected 2013 (dry year), 2015 (wet year) and 2017 (normal year) as the typical hydrological years.

# 3.2. Datasets

(1) Spatial data (i.e., DEM, land use and soil) for model construction and scenario setting (Table 1).

Table 1

Туре	Indicator	Time series	Spatial and temporal resolution	Source	Model	Usage
Land use	Land use type	2015	30 × 30 m	GLC_FCS30-2015	SWATNDP	Input data
DEM	_	_	90 × 90 m	SRTM DEM	SWAT	Input data
Soil	Soil type	_	1:100 million	HWSD	SWAT	Input data
Meteorology	Pr, $T_{Max}$ , $T_{Min}$ , $T_{Ave}$ , Wet, WS and $H_{Sun}$	2009– 2017	Daily	CMDC	SWATNDP	Forcing data
Hydrology	Q	2010– 2012	Daily	Hydrological station (Hengtangcun)	SWAT	Calibration data
	WL	2015– 2016	Daily	Water level logger	NDP	
	Irrigation and flood drainage	2009– 2017	Daily	Surveying	NDP	Forcing data
Water quality	TN, TDN, NO, NH	2010– 2012	Monthly	Water quality monitoring station (Hengtangcun)	SWAT	Calibration data
		2014– 2016	Monthly	Water sampling	NDP	Input data, calibration data
Fertilization	Fertilization amount, Fertilization date	-	-	Surveying	SWATNDP	Input data

Data List of Hydrological and N Transport Models for the Hill and Plain Agricultural Watersheds

*Note. Pr*: daily precipitation (mm);  $T_{Max}$ ,  $T_{Min}$  and  $T_{Ave}$ : daily maximum, minimum and average air temperature, respectively (°C); *Wet*: daily average humidity (%); *WS*: daily average wind speed (m s<sup>-1</sup>);  $H_{Sun}$ : daily sunshine hours (h); *WL*: water level (mm); *TN*: total nitrogen concentration (mg L<sup>-1</sup>); *TDN*: total dissolved nitrogen concentration (mg L<sup>-1</sup>); *NO*: nitrate nitrogen concentration (mg L<sup>-1</sup>); *NH*: ammonia nitrogen concentration (mg L<sup>-1</sup>).

DEM reflected the terrain characteristics of hills (CGIAR-CSI, 2013), therefore it was an important basis for SWAT model to extract water system and to divide sub-watersheds. The land use data was extracted from the GLC\_FCS30-2015 global land-cover products (Liu et al., 2020), which was divided into forest land, grass land, paddy land, dry land, pond and urban land. The soil data was constructed using Harmonized World Soil Database (HWSD, FAO and International, 2019), which were generalized into 6 categories and 9 races, including high active strong acid soil (*Plinthic Alisols, Haplic Alisols, Ferric Alisols*), bottom active strong acid soil (*Humic Acrisols*), embryonic soil (*Ferralic Cambisols, Dystric Cambisols*), high active leaching soil (*Haplic Luvisols*), loose lithologic soil (*Dystric Regosols*) and artificial soil (*Cumulic Anthrosols*).

(2) Meteorological data for model input.

The temperature, solar radiation, wind speed, relative humidity and precipitation were derived from the China Meteorological Data Service Centre (CMDC, 2024). Precipitation had a great impact on runoff, and improving the density of precipitation data was very important for simulation accuracy. Therefore precipitation data of 13 rainfall stations in Tianjintang, Fendai, Hanggai, Shuangshe, Fushi Reservoir, Zhangli, Bingkeng, Laoshikan Reservoir, Xiaofeng, Yinkeng, Licun, Dipu and Hengtangcun were used (Figure 2).

(3) Hydrologic and water quality data for model boundary and model calibration.

The SWAT model was verified by the daily runoff from 2009 to 2012 and the monthly TN load from 2010 to 2012 of Hengtangcun Hydrological Station. The NDP model was verified by the daily water level from 2015 to 2016 measured by the water level recorder (HOBO U20), and the monthly TN concentration from 2014 to 2016 by field sampling and analysis. The drainage and irrigation data of the polders were obtained from water volume and pumping time recorded by the pumping stations.

(4) Agricultural management data for model input.

The agricultural land in our study area was mainly paddy land, and the rest was less than 1/6 of the paddy land. Therefore, only the agricultural management information of paddy land was edited in the models (Table SI-1 in



Supporting Information S1). According to the survey results, the average fertilization rate was set to 50 kg urea/ acre, equivalent to pure N 345 kg/ha (Zhu, 2014).

#### 3.3. Configuration

To develop the SWAT model, the hill agricultural watershed was divided into 45 sub-watersheds (minimum watershed area threshold: 2,000 ha) according to the river network and the outlet; the river network was generated based on DEM; the Hengtangcun Station was defined as the total outlet of the watershed. The 45 sub-watersheds were further divided into 4,993 HRUs according to the consistency of soil type and surface coverage. The information of point source, reservoir, agricultural management (sowing, irrigation, fertilization and crop harvesting) was input according to the actual situation. The water yield, sediment yield and nutrient load on each HRU were calculated separately, furthermore the water balance and total nutrient load of the whole watershed were obtained. The global sensitivity analysis was used to screen the most influential runoff and N-related parameters. Nash-Sutcliffe efficiency coefficient (*NSE*), percentage bias (*PBIAS*) and coefficient of determination ( $R^2$ ) were used to evaluate the goodness of fit of SWAT model for runoff and TN simulation. SWAT model can well capture the seasonal variation trend of discharge and flood peak flow in heavy rainfall events, with the *NSE* between simulated and measured discharge and TN loss were 0.81 and 0.67, respectively. The detailed information of sensitivity parameter analysis processes as well as model calibration and validation processes of SWAT model were given inSection 1 in Supporting Information S1.

The initial conditions of the NDP model included the area, water level, nutrient concentration, population, and number of livestock and poultry of the four land use types in 141 polders of the plain agricultural watershed. The boundary conditions included time series meteorological and irrigation data. NDP model contained five water balance modules composed of 22 parameters and three N dynamic modules composed of 25 parameters. All the parameter values were obtained from previous studies in a typical polder located not far (about 38 km) from the study area (Huang et al., 2018). Since the hydrological and hydraulic conditions of these polders were consistent under the unified management of the Taihu Lake Basin Authority of the Ministry of Water Resources, the simulation results of NDP model were reliable. Compared with the case of watershed simulation using 257 models (Wellen et al., 2015), the simulation results of NDP model were 0.73 and 0.53, respectively; compared with the traditional method, the simulation accuracy by NDP model in the plains was improved by 63%. The detailed information of sensitivity parameter analysis processes as well as model calibration and validation processes of NDP model were given in Section 2 in Supporting Information \$1.

#### 4. Results

#### 4.1. N Loss With Surface Runoff

Among different runoff components, in the hills, surface runoff was the main pathway of TN loss, accounting for 61%. Especially during the rainy season, the dissolved N cannot be fully absorbed by the plant and soil and be partly washed out with the surface runoff. In the dry season, the proportion of TN loss from deep soil water (i.e., interflow or groundwater) was higher (70%) than that from surface runoff (Figure 3a). In the plains, the TN loss from surface runoff was higher (78%) than that in the hills, which should also be concerned in the dry season. Because in the rainy season, the drainage gates were closed to keep the soil water saturation for rice planting. During this period, the water pressure difference between the polders and the surrounding rivers led to an increase in TN loss though seepage. In the dry season, after rice harvest, in order to continue wheat planting, the excess water and N of the paddy land was discharged through drainage. Moreover, due to the slow growth of crops in winter, N fertilization could not be completely absorbed, resulting in excess N being transported to the surrounding rivers through surface runoff (Figure 3b).

#### 4.2. Driving Factors of N Loss With Surface Runoff

Runoff amount were the dominant factor controlling the TN loss. The TN loss in the hills was 3,583.42, 2,209.54 and 1,730.53 ton in the wet year, normal year and dry year, respectively; and the TN loss in the plains was 4,062.55, 2,650.08 and 1,935.57 ton in the wet year, normal year and dry year, respectively. The change trend of TN loss and runoff amount was consistent, showing the characteristics of wet year > normal year > dry year, year, year, we have the the characteristics of wet year > normal year > dry year, year,



# Journal of Geophysical Research: Biogeosciences

10.1029/2023JG007931



Figure 3. TN loss with surface runoff and deep soil water in the hill (a) and plain (b) agricultural watersheds.

properly because N was easy to be accumulated in dry years, and easily to be washed out in wet years (Kaushal et al., 2011).

However, the change trend of TN concentration and runoff amount was not consistent. According to the results of bivariate LISA, in the hills, TN concentration showed a weak negative correlation with runoff amount (Moran's I = -0.293), indicating that due to the dilution of runoff, TN concentration was decreasing. In the plains, there was no significant correlation between the TN concentration and runoff amount (Moran's I = 0.078), that was, the TN concentration did not decrease when the runoff amount was low, which may due to the effect of fertilization at low flow periods (Figures 4b and 4e). The above proved that when the runoff amount was reduced to a certain threshold, the TN concentration was no longer controlled by the runoff amount, but was affected by other factors such as land use and fertilization.

Due to the large amount of N fertilization applied on the agricultural land (i.e., paddy land, dry land and aquaculture pond), it may be the hot spots of TN runoff loss. As shown in Figure 4, the TN concentration increased with the proportion of agricultural land in the hills (Moran's I = 0.356), while the opposite trend was observed in





**Figure 4.** The Spatial autocorrelation analysis of TN loss concentration with surface runoff in the hill (a) and plain (d) agricultural watersheds; the bivariate LISA analysis between TN loss concentration with surface runoff and runoff amount in the hill (b) and plain (e) agricultural watersheds; the bivariate LISA analysis between TN loss concentration with surface runoff and the proportion of agricultural land in the hill (c) and plain (f) agricultural watersheds.

the plains (Moran's I = -0.219). Therefore, agricultural land may be the main N source landscape in the hills, not in the plains. Further analysis showed that the TN loss of paddy land and dry land increased with the runoff amount especially in the hills (Figure 5). Moreover, the TN sink of pond in the plains was much higher than that in the hills, which can compete with the amount of TN source in paddy land and dry land. Therefore, we speculated that the paddy land and dry land may be the main N source in the hills, rather than in the plains.



Figure 5. The linear correlation analysis between TN loss load with surface runoff and runoff amount and the time variation analysis of TN load/sink under different agricultural land types in the hill and plain agricultural watersheds.



**Figure 6.** TN reduction rate under different land and fertilization management scenarios in the hill (a) and plain (b) agricultural watersheds; P2D means from paddy land to dry land; P2U means from paddy land to urban land; P2W means from paddy land to pond; F-20 means 20% reduction in N fertilizer; F-40 means 40% reduction in N fertilizer.

#### 4.3. Contributions of Driving Factors

Multi-scenario simulation results showed that in the hills, the TN loss increased in case of other land use types were converted into paddy land, indicating that paddy land was the main N source landscape and therefore the effect of reducing N fertilization was obvious (Figure 6a). In the plains, the TN loss increased significantly after paddy land converted into urban land, indicating that urban land was the main N source landscape. Moreover, the role of N sink in rural ponds was equivalent to a 20%–40% reduction in N fertilization, with nearly 25% of the watershed TN reduction rate (Figure 6b).

Moreover, it found that the effects of land use transposition and N fertilizer application on N loss were different in wet and dry years. Compared with the normal year and dry year, the N increment effect was doubled in the wet year in the hills (Figure 6a). As the proportion of paddy land of sub-watersheds was increased, the annual TN loss increased faster during high flow periods (Figure 7a). Different from the hills, the N increment effect in the dry year should also be noticed in the plains (Figure 6b). As the proportion of urban land of polders increased, the annual TN loss increased faster during low flow periods (Figure 7b). According to the above results, it can be speculated that natural watersheds were susceptible to agricultural land use and fertilization, especially in high flow periods; complex watersheds (i.e., lowland watershed) had a higher proportion of urban land and more point sources, which can contribute more TN loss in low flow periods.

Further, the pRDA analysis was used to quantify the single and joint contribution of rainfall, land use and N fertilizer application to the total variance of TN loss (Figure 8). TN loss was more sensitive to rainfall than to land



**Figure 7.** The linear correlation analysis between TN loss load with surface runoff and the proportion of paddy land of subwatersheds in the hills (a); the linear correlation analysis between TN loss load with surface runoff and the proportion of urban land of polders in the plains (b).





Figure 8. The pRDA results for relative variation fractions of the N loss with surface runoff explained by rainfall, land use and fertilization in the hill (a) and plain (b) agricultural watersheds.

use and N fertilizer application both in hills and plains. Moreover, the contribution of rainfall in the plains was greater than that in the hills; the contribution of land use in the plains was greater than that in the hills; the contribution of N fertilizer application in the hills was greater than that in the plains.

#### 5. Discussion

#### 5.1. The Response of N Loss to Rainfall, Land Use and Fertilization

Overall, the response of N loss to rainfall was more sensitive than that to land use and fertilization. Rainfall striped N from the soil surface, and runoff provided energy for N to transport downstream. The rainfall intensity may determine the amount of N loss entering the water body, and the rainfall frequency may affect the degree of dilution of the N concentration. Therefore, the N load in the rainy season was higher than that in the dry season, while the N concentration in the dry season was higher than that in the rainy season.

Slope affected the intensity and rate of N loss movement. The generation time of N loss with surface runoff was shorter in the hills than that in the plains after rainfall. Land use directly affected the interaction between runoff and soil, and its changes ultimately caused N loss to a large extent (Tan et al., 2023). The proportion of urban land in the plains increased significantly, and the impervious surface caused serious N loss after rainfall. Moreover, the contribution of land use to N loss in the plains was greater than that in the hills, may because the impact of land use change on surface runoff in the hills was relatively small. The main sources of surface runoff in the hills include rainfall, groundwater and a small proportion of soil water. The groundwater was relatively stable and was mainly affected by climate change (i.e., rainfall) rather than human activities (i.e., land use) in the long-term process (Mani et al., 2016). The soil water saturation of the hills in our study area was high, so the N loss with surface runoff occurred and stopped continuously with the continuation and regression of rainfall, which may to some extent cover up the effects of land use (Chelsea Nagy et al., 2012; Li, Shi, et al., 2020).

The contribution of N fertilization to N loss in the hills was greater than that in the plains. Because paddy land was the most important N source landscape in the hills, fertilization had a greater impact on N loss.



#### 5.2. Implications for N Control

Due to the different N fixation and release capacity in each land use type, there are huge differences in N loss to the surrounding water bodies by different combinations of land use types (Kändler et al., 2017). Hou et al. (2019) found that a decrease of nearly 10% in the N load in surface runoff could be achieved by redistributing land use within the experimental watershed without any changes in any types of land management practices. Therefore, N pollution can be reduced by adjusting land use patterns.

In our study, sub-watersheds with high proportion of paddy land in the hills and polders with high proportion of urban land in the plains are considered to be the key source areas of N loss. Therefore, in the hills, returning agricultural land especially paddy land to forest is necessary. Rapid land cover change as a result of agricultural encroachment onto natural land often lead to long-lasting impairment of soil and water conservation and other crucial ecosystem services. Cebecauer and Hofierka (2008) also found that deforestation and conversion of pastures to agricultural land in the hills was the main factors exacerbating the risk of soil erosion and N loss. In the plains, urban development has accelerated the N loss, so it is necessary to control urban expansion. Roberts et al. (2009) also found that urbanization and increased imperviousness in the future could increase nutrient load to the Chesapeake Bay by 56%, leading to eutrophication and harmful algal blooms. The use of pond systems to reduce watershed N pollution has shown increasing advantages. Most of ponds in the hills are natural ponds with high transparency and few aquatic plants, and their N removal is mainly based on sedimentation. Most of the ponds in the plains are aquaculture ponds, receiving and storing runoff and nutrients from paddy land and urban land (Wang et al., 2019), and their N removal is mainly based on the absorption of aquatic plants and other biochemical reactions. It is obvious that the N sink capacity of ponds in the plains is much higher than that in the hills, with TN reduction of 1,169.87 and 19.23 kg·ha<sup>-1</sup> yr<sup>-1</sup>, respectively. Internationally, pond systems have been considered as one of the best management practices (BMPs) for watershed N control (Li, Liu, et al., 2020).

The N loss entering the water body is promoted by N-fertilization time and rainfall. Rainfall is uncontrollable, while reasonably reducing N-fertilization dose and post-rain fertilization could be implemented based on modern agricultural technology (Ding et al., 2022; Guo et al., 2023; Sun et al., 2012; Zhang, Hou, et al., 2020). Unfortunately, due to higher evapotranspiration and extreme rainfall, higher drought or flood risks would be observed in different regions, which may amplify the effects of land use and fertilization (Aksoy et al., 2021; Amiri & Gocić, 2021; Law et al., 2022). Therefore, more N loss load would be observed in most agricultural land-dominated watersheds. Based on the future climate change, more stringent land management and fertilization control would be needed.

#### 6. Conclusion

This study is comparing the response of N loss with surface runoff to rainfall, land use and fertilization between hill and plain agricultural watersheds using two process-based models. The results showed that among different runoff components, surface runoff was the main pathway of TN loss especially in the plains. The variation trend of TN loss load was consistent with that of runoff amount, showing the characteristics of wet year > normal year > dry year. However, the TN loss concentration was also high during low flow periods, may due to the effects of land use and fertilization. In the hills, paddy land was the main N source landscape compared to the urban land with low population density, whose N increment effect was more obvious in high flow periods. In the plains, urban land with high population density was the main N source landscape compared to the agriculture land with relative natural structure, whose N increment effect was more obvious in low flow periods. The contribution of fertilization to N loss in the hills was greater than that in the plains, because the high proportion of paddy land with abundant N fertilizer demand. Moreover, the contribution of rainfall to N loss in the plains was greater than that in the hills, because the high proportion of urban land with more impervious surface led to serious N runoff loss.

#### **Data Availability Statement**

Land use data in 2015 in Xitiaoxi River Basin can be obtained from https://doi.org/10.5281/zenodo.3986872. DEM in Xitiaoxi River Basin can be obtained from https://data.tpdc.ac.cn/zh-hans/data/acb49ce8-2bfe-4ab4-97ff-e6e727110703. Soil data in Xitiaoxi River Basin can be obtained from https://www.tpdc.ac.cn/zh-hans/data/acb49ce8-2bfe-4ab4-97ff-e6e727110703. Observed daily meteorological data from 2009 to 2017 in Xitiaoxi River Basin were collected from https://doi.org/10.11888/AtmosPhys,tpe.00000049.file. Model configuration,



Acknowledgments

The project was financially supported by National Key Research and Development Program of China (2023YFC3208901, 2023YFC3208903), National Natural Science Foundation of China (52209065, 52121006), Natural Science Foundation of Jiangsu Province of China (BK20220207) and Jiangsu Funding Program for Excellent Postdoctoral Talent (2022ZB439).

### References

6084/m9.figshare.25138838.v1.

Aksoy, H., Cetin, M., Eris, E., Burgan, H. I., Cavus, Y., Yildirim, I., & Sivapalan, M. (2021). Critical drought intensity-duration-frequency curves based on total probability theorem-coupled frequency analysis. *Hydrological Sciences Journal*, 66(8), 1337–1358. https://doi.org/10.1080/ 02626667.2021.1934473

calibration and validation, input and output files in the present study are available online at https://doi.org/10. 6084/m9.figshare.25138847.v1. The NDP model source code is available to download from https://doi.org/10.

- Alamdari, N., Claggett, P., Sample, D. J., Easton, Z. M., & Nayeb Yazdi, M. (2022). Evaluating the joint effects of climate and land use change on runoff and pollutant loading in a rapidly developing watershed. *Journal of Cleaner Production*, 330, 129953. https://doi.org/10.1016/j.jclepro. 2021.129953
- Alldred, M., & Baines, S. B. (2016). Effects of wetland plants on denitrification rates: A meta-analysis. *Ecological Applications*, 26(3), 676–685. https://doi.org/10.1890/14-1525
- Amiri, M. A., & Gocić, M. (2021). Analyzing the applicability of some precipitation concentration indices over Serbia. *Theoretical and Applied Climatology*, 146(1–2), 645–656. https://doi.org/10.1007/s00704-021-03743-5
- Anselin, L. (2004). GeoDa 0.95i Release Notes [Software]. Spatial Analysis Laboratory (SAL). Department of Agricultural and Consumer Economics, University of Illinois, Urbana-Champaign, IL
- Arnold, J. (1994). SWAT Soil and Water Assessment Tool. [Software]. Food and Agriculture Organization of the United Nations. https:// creativecommons.org/publicdomain/zero/1.0/
- Bales, R. C., Molotch, N. P., Painter, T. H., Dettinger, M. D., Rice, R., & Dozier, J. (2006). Mountain hydrology of the western United States. Water Resources Research, 42(8). https://doi.org/10.1029/2005WR004387
- Cebecauer, T., & Hofierka, J. (2008). The consequences of land-cover changes on soil erosion distribution in Slovakia. *Geomorphology*, 98(3), 187–198. https://doi.org/10.1016/j.geomorph.2006.12.035
- CGIAR-CSI. (2013). SRTM DEM dataset in China (2000) [Dataset]. National Tibetan Plateau/Third Pole Environment Data Center
- Chelsea Nagy, R., Graeme Lockaby, B., Kalin, L., & Anderson, C. (2012). Effects of urbanization on stream hydrology and water quality: The Florida Gulf Coast. *Hydrological Processes*, 26(13), 2019–2030. https://doi.org/10.1002/hyp.8336
- CMDC. (2024). Daily Timed Data from automated weather stations in China [Dataset]. China Meteorological Data Service Centre
- Dalu, T., Wasserman, R. J., Magoro, M. L., Mwedzi, T., Froneman, P. W., & Weyl, O. L. F. (2017). Variation partitioning of benthic diatom community matrices: Effects of multiple variables on benthic diatom communities in an Austral temperate river system. Science of the Total Environment, 601–602, 73–82. https://doi.org/10.1016/j.scitotenv.2017.05.162
- de Jong, C., Whelan, F., & Messerli, B. (2005). The importance of a hydrological research framework for water balance studies in mountain basins. *Hydrological Processes*, 19(12), 2323–2328. https://doi.org/10.1002/hyp.5886
- Ding, N., Tao, F., & Chen, Y. (2022). Effects of climate change, crop planting structure, and agricultural management on runoff, sediment, nitrogen and phosphorus losses in the Hai-River Basin since the 1980s. *Journal of Cleaner Production*, 359, 132066. https://doi.org/10.1016/j. jclepro.2022.132066
- FAO & International, I. (2019). China soil map based harmonized world soil database (HWSD) (v1.1) (2009) [Dataset]. National Tibetan Plateau / Third Pole Environment Data Center
- Gabriel, M., Knightes, C., Cooter, E., & Dennis, R. (2018). Modeling the combined effects of changing land cover, climate, and atmospheric deposition on nitrogen transport in the Neuse River Basin. *Journal of Hydrology: Regional Studies*, 18, 68–79. https://doi.org/10.1016/j.ejrh. 2018.05.004
- Guo, H., Li, Y., Wang, X., Ruan, H., Abegunrin, T. P., Wei, L., et al. (2023). Characteristics of nitrogen output during typical rainfall in different sugarcane growth stages in a southern subtropical watershed. Agriculture, 13(8), 1613. https://doi.org/10.3390/agriculture13081613
- Hou, C., Chu, M. L., Guzman, J. A., Acero Triana, J. S., Moriasi, D. N., & Steiner, J. L. (2019). Field scale nitrogen load in surface runoff: Impacts of management practices and changing climate. *Journal of Environmental Management*, 249, 109327. https://doi.org/10.1016/j.jenvman.2019. 109327
- Huang, J., Arhonditsis, G. B., Gao, J., Kim, D.-K., & Dong, F. (2018). Towards the development of a modeling framework to track nitrogen export from lowland artificial watersheds (polders). Water Research, 133, 319–337. https://doi.org/10.1016/j.watres.2018.01.011
- Kändler, M., Blechinger, K., Seidler, C., Pavlů, V., Šanda, M., Dostál, T., et al. (2017). Impact of land use on water quality in the upper Nisa catchment in the Czech Republic and in Germany. *Science of the Total Environment*, 586, 1316–1325. https://doi.org/10.1016/j.scitotenv.2016. 10.221
- Kaushal, S. S., Groffman, P. M., Band, L. E., Elliott, E. M., Shields, C. A., & Kendall, C. (2011). Tracking nonpoint source nitrogen pollution in human-impacted watersheds. *Environmental Science and Technology*, 45(19), 8225–8232. https://doi.org/10.1021/es200779e
- Law, J. Y., Long, L. A., Kaleita, A., Helmers, M., Brendel, C., van der WoudeSoupirWoude, K. M. K., & Soupir, M. (2022). Stacked conservation practices reduce nitrogen loss: A paired watershed study. *Journal of Environmental Management*, 302, 114053. https://doi.org/10.1016/j. jenvman.2021.114053
- Li, B., Shi, X., Lian, L., Chen, Y., Chen, Z., & Sun, X. (2020). Quantifying the effects of climate variability, direct and indirect land use change, and human activities on runoff. Journal of Hydrology, 584, 124684. https://doi.org/10.1016/j.jhydrol.2020.124684
- Li, S., Liu, H., Zhang, L., Li, X., Wang, H., Zhuang, Y., et al. (2020). Potential nutrient removal function of naturally existed ditches and ponds in paddy regions: Prospect of enhancing water quality by irrigation and drainage management. *Science of the Total Environment*, 718, 137418. https://doi.org/10.1016/j.scitotenv.2020.137418
- Li, Y., Wang, H., Deng, Y., Liang, D., Li, Y., & Shen, Z. (2022). How climate change and land-use evolution relates to the non-point source pollution in a typical watershed of China. Science of the Total Environment, 839, 156375. https://doi.org/10.1016/j.scitotenv.2022.156375
- Li, Z., Luo, C., Jiang, K., Wan, R., & Li, H. (2017). Comprehensive performance evaluation for hydrological and nutrients simulation using the hydrological simulation program–Fortran in a Mesoscale Monsoon Watershed, China. *International Journal of Environmental Research and Public Health*, 14(12), 1599. https://doi.org/10.3390/ijerph14121599
- Liu, L., Zhang, X., Chen, X., Gao, Y., & Mi, J. (2020). GLC\_FCS30: Global land-cover product with fine classification system at 30 m using timeseries Landsat imagery [Dataset]. Zenodo. State Key Laboratory of Remote Sensing Science, Institute of Remote Sensing and Digital Earth, Chinese Academy of Sciences. https://doi.org/10.5281/zenodo.3986872



Malago, A., Bouraoui, F., Vigiak, O., Grizzetti, B., & Pastori, M. (2017). Modelling water and nutrient fluxes in the Danube River Basin with SWAT. *Science of the Total Environment*, 603, 196–218. https://doi.org/10.1016/j.scitotenv.2017.05.242

Mani, A., Tsai, F. T. C., Kao, S.-C., Naz, B. S., Ashfaq, M., & Rastogi, D. (2016). Conjunctive management of surface and groundwater resources under projected future climate change scenarios. *Journal of Hydrology*, 540, 397–411. https://doi.org/10.1016/j.jhydrol.2016.06.021

- Oduor, B. O., Campo-Bescós, M. Á., Lana-Renault, N., & Casalí, J. (2023). Effects of climate change on streamflow and nitrate pollution in an agricultural Mediterranean watershed in Northern Spain. Agricultural Water Management, 285, 108378. https://doi.org/10.1016/j.agwat.2023. 108378
- Roberts, A. D., Prince, S. D., Jantz, C. A., & Goetz, S. J. (2009). Effects of projected future urban land cover on nitrogen and phosphorus runoff to Chesapeake Bay. *Ecological Engineering*, 35(12), 1758–1772. https://doi.org/10.1016/j.ecoleng.2009.09.001
- Song, W., Wang, C., Chen, W., Zhang, X., Li, H., & Li, J. (2020). Unlocking the spatial heterogeneous relationship between Per Capita GDP and nearby air quality using bivariate local indicator of spatial association. *Resources, Conservation and Recycling*, 160, 104880. https://doi.org/10. 1016/j.resconrec.2020.104880
- Sun, B., Zhang, L., Yang, L., Zhang, F., Norse, D., & Zhu, Z. (2012). Agricultural non-point source pollution in China: Causes and mitigation measures. Ambio, 41(4), 370–379. https://doi.org/10.1007/s13280-012-0249-6
- Tan, S., Zhao, G., Peng, C., Ye, W., Xie, D., Chen, F., et al. (2023). Multi-scale effects of landscape on nitrogen (N) and phosphorus (P) in a subtropical agricultural watershed: A case of Qi River Basin (QRB), China. *Ecological Indicators*, 147, 110017. https://doi.org/10.1016/j. ecolind.2023.110017
- ter Braak, C. J., & Šmilauer, P. (2012). Canoco reference manual and user's guide: Software for ordination. version 5.0 [Software].
- Vystavna, Y., Paule-Mercado, M. C., Schmidt, S. I., Hejzlar, J., Porcal, P., & Matiatos, I. (2023). Nutrient dynamics in temperate European catchments of different land use under changing climate. *Journal of Hydrology: Regional Studies*, 45, 101288. https://doi.org/10.1016/j.ejrh. 2022.101288
- Wade, A. J., Skeffington, R. A., Couture, R.-M., Erlandsson Lampa, M., Groot, S., Halliday, S. J., et al. (2022). Land use change to reduce freshwater nitrogen and phosphorus will be effective even with projected climate change. *Water*, 14(5), 829. https://doi.org/10.3390/ w14050829
- Wang, Y., Sun, B., Gao, X., & Li, N. (2019). Development and evaluation of a process-based model to assess nutrient removal in floating treatment wetlands. Science of the Total Environment, 694, 133633. https://doi.org/10.1016/j.scitotenv.2019.133633
- Wellen, C., Kamran-Disfani, A. R., & Arhonditsis, G. B. (2015). Evaluation of the current state of distributed watershed nutrient water quality modeling. *Environmental Science and Technology*, 49(6), 3278–3290. https://doi.org/10.1021/es5049557
- Wenng, H., Bechmann, M., Krogstad, T., & Skarbøvik, E. (2020). Climate effects on land management and stream nitrogen concentrations in small agricultural catchments in Norway. Ambio, 49(11), 1747–1758. https://doi.org/10.1007/s13280-020-01359-z
- Xue, L., Yu, Y., & Yang, L. (2011). Nitrogen balance and environmental impact of paddy field under different N management methods in Taihu Lake region. *Environmental Sciences*, 32(4), 1133–1138. https://doi.org/10.1016/S1671-2927(11)60313-1
- Yuan, N., Li, Y., Xiong, Y., Xu, B., Liu, F., & Fu, H. (2023). Scale effects on the reduction of drainage water and nitrogen and phosphorus loads in hilly irrigation areas. *Agronomy*, 13(8), 2083. https://doi.org/10.3390/agronomy13082083
- Zhang, S., Hou, X., Wu, C., & Zhang, C. (2020). Impacts of climate and planting structure changes on watershed runoff and nitrogen and phosphorus loss. *Science of the Total Environment*, 706, 134489. https://doi.org/10.1016/j.scitotenv.2019.134489
- Zhang, W., Li, H., Pueppke, S. G., Diao, Y., Nie, X., Geng, J., et al. (2020). Nutrient loss is sensitive to land cover changes and slope gradients of agricultural hillsides: Evidence from four contrasting pond systems in a hilly catchment. Agricultural Water Management, 237, 106165. https:// doi.org/10.1016/j.agwat.2020.106165
- Zhao, C. S., Yang, Y., Yang, S. T., Xiang, H., Wang, F., Chen, X., et al. (2019). Impact of spatial variations in water quality and hydrological factors on the food-web structure in urban aquatic environments. *Water Research*, 153, 121–133. https://doi.org/10.1016/j.watres.2019.01.015
- Zhou, F., Xu, Y., Chen, Y., Xu, C. Y., Gao, Y., & Du, J. (2013). Hydrological response to urbanization at different spatio-temporal scales simulated by coupling of CLUE-S and the SWAT model in the Yangtze River Delta region. *Journal of Hydrology*, 485, 113–125. https://doi. org/10.1016/j.jhydrol.2012.12.040
- Zhu, X., Chang, K., Cai, W., Zhang, A., Yue, G., & Zhao, X. (2022). Response of runoff and nitrogen loadings to climate and land use changes in the middle Fenhe River basin in Northern China. *Journal of Water and Climate Change*, 13(7), 2817–2836. https://doi.org/10.2166/wcc. 2022.121

Zhu, Y. (2014). SWAT model based analysis on N/P loss loadings from a mountainous watershed. Zhejiang University. (in Chinese).